

AIMS Energy, 13(1): 188–230. DOI: 10.3934/energy.2025007 Received: 14 September 2024 Revised: 29 December 2024 Accepted: 23 January 2025 Published: 28 February 2025

https://www.aimspress.com/journal/energy

Review

Cutting-edge progress in offshore wind and tidal stream power technology—State-of-the-Art

Ladislas Mutunda Kangaji^{1,*}, Atanda Raji¹ and Efe Orumwense²

- ¹ Department of Electrical and Electronic Engineering, Peninsula University of Technology, Cape Town, 75352, South Africa
- ² Department of Mechanical Engineering, Peninsula University of Technology, Cape Town, 75352, South Africa
- * Correspondence: Email: Laskangaji@gmail.com; Tel: +27733177145.

Abstract: The growing global demand for clean and sustainable energy has underscored the critical importance of offshore wind and tidal stream power technologies in addressing energy and environmental challenges. While these technologies hold significant potential to harness marine energy resources, gaps remain in understanding their readiness levels, integration potential, and pathways to overcome existing barriers. This study aims to bridge these gaps by providing a comprehensive review of the latest advancements in offshore wind and tidal stream power systems, focusing on innovations in turbine design, materials, and hybrid systems that combine wind and tidal energy. The methodology involves an extensive review and synthesis of recent research, project reports, and industry developments to evaluate the current technological state, challenges, and opportunities. Key findings include notable progress in turbine efficiency and hybrid system integration, which collectively improve energy conversion efficiency, scalability, and reliability. However, the study identifies persistent barriers such as high costs, environmental impact, and competition with more established renewable energy sources like solar and onshore wind. This paper emphasizes the importance of hybrid systems as a transformative approach to maximizing marine resource utilization and enhancing energy supply stability. The findings have significant implications for guiding future research, fostering innovation, and informing investment strategies in the marine renewable energy sector.

Keywords: hybrid wind-tidal energy systems; marine renewable energy; voltage source converter (VSC); tidal power integration

1. Introduction

The growing demand for electricity and the urgent need to reduce carbon emissions have intensified global efforts to develop renewable energy sources. Offshore energy systems, including wind, ocean currents, and wave energy, are key contributors to these efforts [1]. Offshore wind energy, for example, demonstrates 48% greater sustainability than onshore wind, according to life cycle analysis (LCA) [2]. However, the high cost of electricity from offshore wind remains a major obstacle to widespread adoption [3]. Ocean renewable energy (ORE) technologies, such as wave energy, tidal stream energy, and ocean flow energy, are gaining attention as viable additions to the renewable energy mix [4]. Emerging technologies such as salinity gradient engines and ocean thermal energy conversion (OTEC) also show promise [5]. Tidal energy stands out for its predictability and significant potential, with an estimated global capacity of 3000 GW, including 1000 GW in shallow waters. By 2021, Europe had installed 30.2 MW of tidal stream systems, indicating growing feasibility and scalability [6].

Although less advanced than offshore wind or wave energy, tidal stream systems have the potential to produce around 20 TWh per year [7]. Nevertheless, tidal stream and offshore wind systems have not yet reached full economic viability. At £140/MWh for offshore wind and £305/MWh for tidal stream energy, electricity strike costs are still expensive [8]. The UK, with its vast tidal energy resources, could generate over 4 TWh annually using current technology, with future projections reaching 94 TWh per year and an installed capacity of 36 GW [9,10].

Ocean energy (OE), or "Blue Energy", encompasses power derived from waves, currents, thermal gradients, and salinity gradients [11]. Among these, offshore wind energy is the most advanced and commercially viable. It is considered the most promising marine renewable energy (MRE) category for large-scale maritime applications. Tidal current energy converters (TCECs), including horizontal-axis turbines, vertical-axis turbines, and oscillating hydrofoils, have seen advancements in blade design, control strategies, array configurations, and environmental impact studies [7,12].

The technology readiness level (TRL) of tidal energy has progressed from laboratory-scale development to the pre-commercial stage [13]. Offshore energy systems include current turbines, wave energy converters, tidal stream generators, and offshore wind turbines, with tidal stream energy being more predictable due to less variability over extended periods. These systems are crucial for maritime electrification and military operations, though research has often focused on specific applications rather than holistic integration [14].

Combining multiple renewable energy sources with storage creates a reliable, eco-friendly, cost-effective energy solution. The integration of wind and tidal energy, in particular, presents a significant opportunity, but this requires collaborative research efforts to develop efficient and reliable systems [15,16]. This paper reviews recent advancements in offshore wind and tidal stream technologies, focusing on innovations that improve energy efficiency, scalability, and reliability. It also highlights ongoing challenges, such as cost competitiveness, and discusses future trends that could drive a wider adoption of these renewable energy solutions.

Section 2 provides an overview of energy policies and underscores the importance of electrical generators as key components of ocean energy converters. Section 3 examines four configurations of low-voltage (LV) converters commonly used in marine energy systems. Section 4 focuses on using synchronous generators in tidal current energy converters (TCECs) and classifies marine current turbines based on their interaction with water motion, noting areas where categories overlap.

Section 5 concludes by exploring hybrid configurations that combine offshore wind and tidal stream technologies. These configurations have the potential to maximize resource utilization and address energy supply challenges. This review aims to help researchers, policymakers, and industry stakeholders better understand the current state and future potential of offshore renewable energy systems. Unlike previous studies that focus on offshore wind or tidal energy individually, this review highlights the integration of hybrid systems that combine these technologies. It examines how the synergy between wind and tidal power can optimize resource use, enhance energy stability, and tackle cost and environmental challenges. By focusing on these hybrid solutions, the study provides practical insights to drive the transition to more sustainable offshore energy systems.

2. Electrical generators are crucial components in ocean energy converters

2.1. Introduction

The European Green Deal prioritizes energy transition to achieve climate neutrality by 2050, while also addressing biodiversity loss and pollution. To meet energy and climate goals, the EU has set 2030 targets to boost renewable energy use [11].

The vast potential of oceanic energy resources has long been overlooked. However, a significant shift is occurring. There is now a growing focus on harnessing eco-friendly energy from the ocean, marking an important development in the energy sector [15,16]. This shift is driven by the belief that wind and tidal energy could significantly contribute to meeting global electricity demands [17]. Astonishingly, the combined energy from wind and tides exceeds current global energy consumption by a factor of 300. However, despite this immense capacity, these resources have remained largely unused, as shown in Figure 1 [18]. This work explores tidal stream energy fundamentals and turbine classifications [19]. However, the current state of marine energy has not yet met the criteria outlined in the Sustainable Development Scenario (SDS), which mandates an annual growth rate of 23% until 2030. On the other hand, countries such as Canada, the UK, China, and Australia have advanced marine energy projects from 10 kW to 1 MW (IEA 2020b) [19].



Figure 1. Cumulative global capacity of offshore energy from 2016 to 2022 [20].

Renewable marine energies include wind, wave, tidal, and ocean currents and innovative approaches like ocean thermal energy conversion (OTEC) and osmotic power, which harness energy from temperature variations and salinity gradients, respectively [18]. Evidence of wind energy's expansion manifests in installed capacity statistics. China leads the pack, boasting a substantial 237 GW of wind power, closely trailed by the US with approximately 105 GW [21]. Europe remarkable progress, with installed capacity soaring from 1.48 GW in 2008 to a formidable 25 GW in 2020. Offshore renewable energy (ORE) is a type of renewable energy that can be harnessed from various sources, including wind (both bottom-fixed and floating), ocean energy (tidal and wave), and floating solar technologies. These technologies are currently at different stages of development (see Figure 2) [21].



Figure 2. Overview of ORE technologies [18].

2.2. Comparisons between onshore and offshore power systems

The term *offshore*, particularly in the context of tidal stream power, has multiple interpretations that need clarification in this review [22].

As defined by the International Convention for the Safety of Life at Sea, offshore includes internal waters, territorial seas, contiguous zones, exclusive economic zones, and international waters. For this review, offshore power systems (OffPS) are defined as systems located in water, submerged beneath saline layers, or positioned deep within the subsea. While some principles might apply to freshwater environments like rivers and lakes, the focus remains on marine, subsea, deep-sea, offshore islands, and maritime environments [23].

Although OffPS shares some similarities with land-based power systems, they encounter unique technical challenges and risks. Figure 3 illustrates the key factors that shape the distinct resilience and reliability of offshore power system components. Offshore systems, especially those on floating platforms, must withstand oscillations from waves, exposure to moisture, and the corrosive effects of saline water [24]. Onshore high-voltage transmission towers are unsuitable for these conditions due to the harshness of the deep-sea environment. Additionally, subsea cable installation differs significantly from that of overhead land-based cables, with the added benefit that subsea lines are immune to lightning strikes. Consequently, the interaction between the subsea ecosystem and offshore power systems differs substantially from onshore systems [25].



Figure 3. Contrasts between onshore and offshore power systems in terms of transmission [26].

2.2.1. Generalized offshore power system architecture

A comprehensive offshore power system (OffPS) framework integrates several interconnected subsystems at sea, as shown in Figure 4. These subsystems include power generation, transmission, distribution, consumption, energy storage, intelligence, and environmental considerations [11]. The generalized architecture in Figure 4 serves as a reference for this research, organizing OffPS into seven key components: power generation, transmission, distribution, consumption, energy storage, offshore intelligence, and environmental factors. Each plays a crucial role in managing the flow of electricity from production to transmission, storage, and conversion [27]. Beyond wind power, tidal stream energy is increasingly vital to OffPS, particularly in areas with strong tidal currents. Notable examples include the Pentland Firth in Scotland and South Africa's Agulhas Current along the Western Cape. These locations benefit from predictable and reliable tidal cycles, offering a consistent energy source that complements the variability of wind and tidal power [28]. This creates a more balanced and resilient renewable energy mix. Energy storage has gained prominence as it mitigates the intermittency of wind and tidal energy sources [24]. Larger wind turbines also help by capturing more energy, reducing overall installation and maintenance costs compared to smaller units. Wind turbines are typically installed across vast areas, forming wind farms connected to national grids. These farms can be located onshore or offshore, though offshore installations often involve higher initial and maintenance costs due to the need for robust foundations and submarine cables. Similarly, offshore tidal turbines face challenges from underwater currents and corrosive seawater, requiring durable construction [29]. While onshore wind farms are more accessible and cost-effective, offshore wind and tidal farms demand advanced transmission technologies. These include series/parallel AC/DC configurations as well as high-voltage AC (HVAC) and high-voltage DC (HVDC) systems, essential for connecting offshore systems to the grid. However, the rapid integration of wind and tidal energy into electrical grids raises concerns about stability, security, and efficiency. As a result, many countries have updated grid codes to govern the connection of large-scale renewable energy projects [30].

The offshore intelligence subsystem plays a critical role in managing, maintaining, and automating OffPS by processing data from all other subsystems, ensuring smooth operations and reliability [18]. The environmental subsystem interacts with natural forces such as weather, hurricanes, tsunamis, and powerful tidal currents, as well as artificial elements like vibration-isolation systems and submarine hulls that protect against deep-sea pressures. These structures also serve as ground terminals and voltage references for OffPS grounding systems [31]. Internal interactions within OffPS form a complex network of data and power connections. Figure 4 illustrates these interactions, with dashed lines representing data flows and solid lines indicating power connections [32]. This conceptual network does not correspond directly to physical systems but provides a broad depiction of how offshore systems function, incorporating the diverse capabilities of wind and tidal energy installations in locations like the Pentland Firth and the Agulhas Current [33].



Figure 4. Overall structure of an offshore power system and its various interconnections depicted in the generalized architecture [24].

The following sections delve into an exhaustive exploration of each subsystem in finer detail. Offshore power generation involves the conversion of energy sources into electricity exclusively within the offshore environment [24]. While several energy sources, such as solar, wind, and fossil fuels, exist in both onshore and offshore contexts, some are unique to the oceans, such as waves and tides [34]. These offshore power generation systems are categorized based on their energy source and can include connections to auxiliary systems such as subsea structures and platforms. In this context, these external systems are considered part of the offshore environment, distinct from the primary power generation, transmission, distribution, consumption, energy storage, and intelligence processes [35].

2.3. Overview of offshore technology in development

The development of offshore renewable energy technologies has primarily focused on designing efficient energy conversion systems to harness the kinetic energy of natural resources like wind and tidal currents, which turn the rotor to drive the generator, transforming tidal kinetic energy into electrical power. Unlike wind turbines, tidal systems must contend with the unique challenges of underwater environments, such as water density resistance and marine corrosion [26]. However, the predictability of tidal cycles provides a stable and reliable energy source, making tidal energy a valuable component of the renewable energy landscape [36]. The following section explores the components and design considerations involved in the power conversion chain for tidal current energy systems.

2.3.1. Power conversion chain design for tidal current energy systems

Tidal stream turbine systems share similarities with wind turbines in both design and operation, allowing for effective comparisons in reliability assessment and drivetrain configurations. Although comprehensive data on tidal stream turbine failures are scarce, the available documented cases provide valuable insights into performance challenges [3]. These turbines are typically classified as either vertical-axis or horizontal-axis, with the latter being more common in tidal energy applications [6].

A horizontal-axis tidal stream turbine consists of key components, including rotor blades, a drivetrain, and a generator. The turbine captures the kinetic energy of tidal currents through its rotor blades, converting the water's momentum into mechanical energy [6]. This mechanical energy is then transferred via the drivetrain to the generator, where it is transformed into electrical power. The energy transfer may occur directly through a direct-drive system or via a gearbox that adjusts rotational speed for optimal power conversion [37].

The tidal current energy conversion (TCEC) system is a sophisticated assembly designed to convert tidal currents into electrical power. This system typically includes a turbine, a pitch control system, and an advanced drivetrain. The drivetrain system, as shown in Figure 5, contains components like the main drive shaft, bearings, gearbox, coupling, brake, and generator, all of which are crucial for efficient energy conversion [37].

A key factor to consider is the density of water, which is approximately 800 times greater than air, making tidal energy systems operate under unique conditions. These systems are generally optimized for flow speeds of approximately 3 m/s, which is the velocity at which the system reaches maximum power output [38]. The design of the turbine blades and pitch control must be tailored to the specific site conditions, including tidal speeds exceeding 2.25 m/s and water depths between 25 and 50 m. For efficient power generation, turbines require large swept areas and strong tidal currents, which are relatively rare and typically found in specific geographic features like narrow straits [38].

Key control mechanisms in tidal energy conversion systems include:

- Blade pitch control: Adjusting the blade pitch allows for precise regulation of power capture and mechanical loads. Hydraulic actuators or servo pitch motors facilitate these adjustments, enabling the turbine to adapt to changing tidal conditions.
- Generator torque control: Managing the generator torque is critical for optimizing the conversion of mechanical energy into electricity, allowing for efficient power generation.
- Yaw control: While more commonly used in wind turbines, yaw control systems in tidal turbines may help align the rotor with tidal currents, enhancing energy capture and overall system efficiency [26].



Figure 5. Power conversion chain of TCEC [39].

Gearbox failures have long been a significant challenge in the energy sector, leading to substantial downtime and repair costs [40]. Although much of the focus has been on wind turbines, tidal stream turbines encounter similar issues related to gearbox reliability [23,41]. Gearbox components are standardized, accessible, and cost-effective, yet they are pivotal to the efficient operation of tidal energy systems. Gearboxes, based on mature technology, are widely used across various sectors, including transportation and manufacturing. These components are essential and typically exhibit a higher mean time to repair (MTTR) compared to other subsystems [41]. The extensive theoretical and practical knowledge available for geared systems offers valuable insights for designing geared tidal turbines. By integrating advancements in gearbox design, condition monitoring, and maintenance practices, it is possible to significantly improve the reliability of tidal stream systems and mitigate potential failures [42]. Optimizing tidal stream energy systems requires leveraging these advancements in gearbox technology and maintenance strategies. Such improvements will enhance system performance and reduce downtime, thereby increasing the overall efficiency and reliability of tidal current energy systems [43].

At the core of the power conversion process lies the mechanical transmission system, which facilitates the transfer of kinetic energy from the tidal current to the generator. The design and configuration of this system directly impact the efficiency and reliability of tidal energy conversion. The next section delves into the key role of mechanical transmission, including gearbox designs and alternatives such as direct-drive systems, which are increasingly favored in offshore applications.

2.3.2. Mechanical transmission

In tidal stream energy systems, the gearbox is a crucial component of the mechanical transmission system. Tidal stream turbines typically operate at low rotor speeds due to the slow movement of tidal currents [44]. To convert this low-speed rotational energy into higher speeds required for electricity generation, a gearbox is used. The gearbox increases the rotor's speed from the low revolutions per minute (rpms) generated by the tidal currents to the higher rpms needed by the generator [45]. In some tidal stream systems, particularly those utilizing multi-pole synchronous generators connected to the

grid via a power converter, the gearbox may be omitted. These generators are capable of operating efficiently at lower speeds, making a gearless design feasible. This approach is particularly beneficial for offshore applications, where minimizing maintenance is essential due to the challenging environment [46]. By eliminating the gearbox, the system reduces mechanical complexity and maintenance requirements [47]. For tidal turbines with smaller rotor diameters, gearboxes may also be unnecessary. Smaller rotors can achieve higher rotational speeds given the tidal currents, which may directly match the generator's requirements. This enables the use of a direct-drive system, simplifying the mechanical design and reducing maintenance needs. Direct-drive tidal stream turbines are increasingly used in standalone DC applications and battery charging systems [48].

The nacelle houses all critical equipment for the tidal turbine, including the drivetrain components (rotor shaft, gearbox, and bearings), electrical generator, power electronics, and control systems. It connects to the support structure via bearings, allowing for rotation to maintain optimal alignment with the tidal currents. The strategic placement of the drivetrain and associated components within the nacelle ensures efficient operation and facilitates maintenance [49].

2.3.3. Hydrodynamic model and maximum power extraction

Tidal stream turbines (TSTs) harness energy from tidal currents, which are driven primarily by the gravitational forces of the moon and sun. Unlike wind, tidal currents are more consistent and predictable. To understand energy extraction from tidal streams, consider the flow of water through the rotor disk, which represents the swept area of the turbine [34]. The mass flow rate of water $\left(\frac{dm}{dt}\right)$ through the rotor disk is dependent on the water density $[\rho]$ and the flow velocity U [m/s]. For a uniform flow velocity across the rotor's swept area (A), the mass flow rate is given by [44]:

$$\frac{dm}{dt} = \rho. A. U \tag{1}$$

The instantaneous kinetic power available from the tidal stream is calculated using:

$$P = \frac{1}{2}\rho. A. V^{3}_{flow}$$
(2)

where V_{flow} is the free stream velocity of the tidal current. TSTs cannot extract all the available power due to the need for some residual flow behind the rotor. The theoretical maximum for power extraction is governed by the Betz limit [50].

The primary objective of TSTs is to maximize power extraction from tidal streams. The power delivered to the rotor and the torque developed can be expressed as follows:

$$P = T\omega$$
(3)

where ω represents the angular velocity of the turbine and T is the torque on the shaft. The angular velocity can be determined by:

$$\lambda = \frac{\omega \mathbf{r}}{\mathbf{v}} \tag{4}$$

where r is the turbine radius, λ is the tip-speed ratio, and V_{flow} is the stream velocity. The torque on

the turbine shaft is given by:

$$T = \frac{1}{2}\rho chr V_{rel}{}^2 C_T$$
(5)

where C is the blade chord length, V_{rel} is the relative velocity, and C_T is the torque coefficient [34]. Operating the turbine at variable speeds allows for optimal power extraction across a range of tidal flow speeds [49]. Figure 6 shows the characteristic curve for optimal power extraction and illustrates how the turbine's performance aligns with varying tidal flow speeds [51].



Figure 6. Power characteristics of tidal turbine [49].

The next section presents a hydrodynamic model and discusses strategies for maximizing power extraction from tidal currents.

2.3.4. Tidal stream turbines and offshore variable-speed wind turbines

The recent development of larger tidal stream system turbines has led to a shift from fixed-speed to variable-speed operation. This shift addresses grid code requirements and reduces mechanical loads [50]. Figures 11 and 12 present configurations employing a permanent magnet synchronous generator (PMSG) and a full power converter (FPC), relevant to TSTs [41,52]. By routing all turbine power through converters, the dynamic operation of the PMSG is decoupled from grid frequency variations, ensuring stable grid operation despite fluctuations in tidal stream speed [53]. While the hydrodynamic model focuses on maximizing power extraction from tidal currents, integrating this energy into the grid requires additional components, such as voltage transformers. These transformers ensure that the power output meets the required voltage levels for safe and efficient transmission. The next section examines the role of voltage transformers in tidal energy systems and their significance in grid integration.

The voltage transformer adjusts the turbine's internal voltage, typically between 690 and 1000 V, to match the collector grid voltage, commonly rated at 33 or 36 kV. This transformation incurs power losses that are converted into heat, necessitating dissipation via the cooling system [53]. Many tidal stream turbine systems neglect to fully account for these transformer losses in their control and management systems. Recent developments show that newer and larger turbines are incorporating advanced controllers to mitigate these losses [41,52]. While the voltage transformer plays a critical role in adjusting turbine output to match grid voltage, ensuring optimal energy capture and conversion requires additional subsystems for efficient turbine operation. Among these, the yaw drive is essential for maintaining the turbine's alignment with changing tidal or wind directions, thereby maximizing energy generation. The following section explores the functionality and importance of yaw drive systems in renewable energy technologies.

2.4.1. Yaw drive

The yaw drive is vital in modern horizontal-axis wind turbine yaw systems. It actively adjusts to keep the rotor aligned with changing wind directions, optimizing energy production. A wind vane measures wind direction at the nacelle's rear [53]. Yaw angle errors lead to reduced energy capture, with power losses proportional to the cosine of the yaw error [53,54]. In addition to optimizing turbine alignment, safety mechanisms like the mechanical brake are integral to the reliable operation of tidal stream turbines, especially during emergencies or maintenance. This section examines the mechanical brake's role in ensuring system stability and safety.

2.4.2. Mechanical brake

The mechanical brake is crucial in tidal stream turbines, serving as an emergency backup if other braking systems fail. Positioned between the gearbox and the generator, it also prevents unexpected turbine activation during maintenance. The mechanical brake engages after other braking methods have reduced the turbine speed, minimizing wear [52]. The following section discusses advancements in power electronic converters and their critical role in tidal energy systems.

2.4.3. Power electronic converter

Advancements in power electronics have significantly enhanced the operation of renewable energy [52]. Power electronic converters are essential for interfacing tidal stream and wind turbines with the grid. As tidal turbines operate at variable speeds, their generator's electrical frequency fluctuates, requiring decoupling from the grid frequency via these converters [51]. This decoupling enables turbines to operate efficiently across varying speeds, enhancing power capture compared to fixed-speed models. Common configurations include gear-coupled double-fed induction generators (DFIGs), where power electronics manage rotor voltage and frequency, or direct-coupled synchronous generators (SGs), with power electronics controlling stator voltage and frequency. Both configurations may use brushless field excitation or permanent magnet excitation [47].

Effective variable-speed operation requires a control system tailored to the specific tidal stream turbine, ensuring maximum power extraction while maintaining stable grid voltage and frequency [54]. As switching frequencies increase to develop more compact and efficient equipment, emissions in the 2–150 kHz frequency spectrum, known as super harmonics, become more pronounced [55]. Key power converters include inverters, rectifiers, DC/DC converters, and DC/AC inverters [56]. Building on the advancements in power electronic converters, integrating these systems into variable-speed turbine configurations has significantly enhanced the efficiency and flexibility of energy systems. The direct-in-line variable-speed turbine with a full-scale power converter represents one such innovative design, addressing key challenges associated with fixed-speed systems. The next section examines the advantages and trade-offs of this configuration in detail.

2.4.4. Direct-in-line variable speed wind turbine with full-scale power converter

In fixed-speed systems, the generator connects directly to the grid (see Figure 9), maintaining a steady rotor rotation determined by grid frequency, gear ratio, and generator pole pairs [16]. Fixed-speed configurations are generally straightforward, reliable, require less maintenance, and are cost-effective [57]. However, they present several drawbacks:

- Limited energy capture: Power extraction depends on the rotational speed of the prime mover and environmental conditions. Variable-speed solutions can improve efficiency by 5%–20% across varying conditions.
- Mechanical stress: Fixed speeds cause torque pulsations, leading to mechanical strain. Systems using squirrel-cage induction generators experience less mechanical stress due to slip compliance during transients.
- Power quality: Mechanical power fluctuations affect both the drive train and the grid, leading to degraded power quality. Fixed-speed systems, particularly those with low-inertia drive trains, can generate noticeable electrical power fluctuations.
- Reactive energy compensation: Squirrel-cage induction generators consume reactive energy, often requiring additional compensation components.

Converters, which serve as interfaces between the load/generator and the grid, are crucial for ensuring reliability, efficiency, and cost-effectiveness. They consist of power electronic devices, driving circuits, protection systems, and control circuits [58].

Currently, two main types of converters are used: grid-commutated and self-commutated converters [59]. Grid-commutated converters, which use thyristors and multiple pulses, generate integer harmonics but lack reactive power control [60]. Variable-speed solutions offer several advantages over fixed-speed systems:

- Decoupling: Power electronics decouple the grid from the generator.
- Enhanced power take-off: Improved alignment between the resource and prime mover speed enhances power take-off.
- Energy storage: The rotor and drive train can act as a flywheel, storing or delivering energy and reducing power fluctuations.
- Reduced stress mechanical stress: Variable-speed systems reduce mechanical stress due to inherent mechanical compliance.
- Full power control: Both active and reactive power can be controlled.

However, variable-speed solutions entail higher costs due to power electronics. Common configurations include fully controllable back-to-back converters in either a double-fed induction generator setup or a full-converter configuration [61]. In the DFIG arrangement shown in Figure 7, the rotor winding connects to the grid via a bidirectional power converter, which typically comprises 30%–40% of the rated power. This configuration enables variable-speed operation and control over active and reactive power within specific limits but requires slip rings, increasing maintenance needs [62].



Figure 7. DFIG topology.

Power converters are critical components in tidal stream energy systems, acting as interfaces between the generator and the grid to manage bidirectional power flow [49]. Key considerations for these converters include reliability, efficiency, and cost. They consist of power electronic devices, drive systems, protection circuits, and control mechanisms [59]. The primary converter types used are grid-commutated and self-commutated converters. Grid-commutated converters, utilizing thyristors, produce integer harmonics and lack reactive power control capabilities [61].

Variable-speed systems present several advantages over fixed-speed configurations in tidal stream turbines:

- Grid decoupling: Power electronics enable the decoupling of the generator from the grid, providing greater operational flexibility.
- Enhanced power take-off (PTO): By aligning turbine speed with tidal flow conditions, power extraction is optimized.
- Energy storage: The rotor and drive train can act as a flywheel, storing or delivering energy to mitigate power fluctuations.
- Reduced mechanical stress: Variable-speed operation reduces mechanical stress due to the inherent mechanical compliance of the system.
- Comprehensive power control: Both active and reactive power can be precisely managed.

Presently, the most common configurations involve employing a fully controllable back-to-back converter, either within a double-fed induction generator setup (refer to Figure 8) or in a full-converter configuration (refer to Figure 9) [61]. Despite these benefits, variable-speed systems incur higher costs due to the advanced power electronics required. Common configurations involve fully controllable back-to-back converters, either in a double-fed induction generator (DFIG) setup or a full-converter

arrangement [63]. In the DFIG configuration, the rotor winding connects to the grid via a bidirectional power converter, typically accounting for 30%–40% of the rated power. This setup facilitates variable-speed operation and enables control over both active and reactive power within specified limits. However, the requirement for slip rings in this arrangement can increase maintenance demands [62].



Figure 8. Full converter topology.



Figure 9. Back-to-back converter.

Power converters are fundamental to the operation of both variable-speed and fixed-speed tidal turbines, acting as interfaces between the generator and the grid. They ensure bidirectional power flow, maintain grid stability, and enable precise control of active and reactive power. In the next chapter, we delve deeper into the design, types, and functionality of converters used in tidal stream energy systems, highlighting their role in improving efficiency and reliability.

3. Converters

This section examines four LV converter configurations, each tailored to specific tidal stream turbine grid connections, typically utilizing standard voltages of 690 and 575 V. Power converters are essential components in tidal stream energy systems, serving as interfaces between the generator and the grid. They enable efficient power conversion, regulate energy flow, and support grid stability in variable operating conditions. The following subsections detail the key converter configurations employed in tidal stream turbines, beginning with the widely adopted full-scale back-to-back two-level voltage source converters (BTB 2L-VSCs).

3.1. Full-scale BTB 2L-VSCs

Figure 10 illustrates a typical Type 4 tidal conversion system (TCS) using full-scale back-to-back (BTB) two-level voltage source converters (2L-VSCs) [63]. These consist of a voltage source rectifier (VSR) and a voltage source inverter (VSI), connected by a DC-link capacitor [64]. Both the VSR and VSI are realized through low-voltage insulated gate bipolar transistors (LV-IGBTs), arranged in a matrix configuration. The DC-link unit, consisting of capacitors arranged in series or parallel, achieves the required voltage and capacitance. This link decouples the generator from the grid, preventing generator transients from impacting grid stability [65].

This mature converter topology is used in approximately 90% of Type 4 tidal stream turbines rated below 0.75 MW and supports a range of tidal stream generators like permanent magnet synchronous generators (PMSG), wound rotor synchronous generators (WRSG), and squirrel cage induction generators (SCIG). The converter's power rating is typically matched to the generator output, so a 0.75 MW generator connects to a 0.75 MW converter. The VSR controls the generator's torque and speed, while the VSI manages the DC-bus voltage and grid reactive power [66].

To minimize switching losses and enhance power density, the VSR and VSI operate with a switching frequency between 1 and 3 kHz. LCL filters are used on the grid side to reduce harmonic distortion in grid currents [67]. Although not shown, generator-side harmonic filters are also required [68]. The entire converter system—filters, rectifiers, DC-link, and inverters—is housed in a cabinet located in the turbine nacelle [69]. After filtering, the AC output is transmitted through cables to a step-up transformer at the tower's base. These low-voltage, high-current AC cables, though costly and prone to transmission losses, benefit from mass production, reducing overall expenses [70].

An example of a widely used commercial converter module is Semikron's SKiiP, known for its compact design and high power density [71]. It integrates heat sinks, semiconductor switches, and gate drivers into a single unit. While full-scale BTB converters are commonly used in small-scale tidal systems for their simplicity and effectiveness, larger systems often employ partial-scale converters to reduce costs and improve flexibility. The next section explores partial-scale BTB 2L-VSC configurations, which are specifically designed for semi-variable speed tidal systems [72].

3.2. Partial-Scale BTB 2L-VSCs

Figure 11 shows a Type 3 tidal conversion system that uses BTB converters in semi-variable speed tidal systems [73]. In this configuration, the stator is directly connected to the grid, while the rotor is linked to the power converter [74]. The converter's power rating is typically around 30% of the generator's total capacity. For example, a 2.5 MW doubly-fed induction generator (DFIG) would use a 0.75 MW converter [75]. The full-scale and partial-scale converters regulate active and reactive power, control DC-link voltage, and maintain grid power factor [76].



Figure 10. Type 4 TCS with two-level BTB voltage source converters (mainstream commercial power converter configuration) [63].



Figure 11. Type 3 TCS with two-level BTB voltage source converters [63].

As the power demands of tidal stream turbines increase, more advanced converter configurations are required. Parallel BTB VSC systems with a common DC-link provide a scalable solution for turbines with power ratings exceeding 0.75 MW.

3.3. Parallel BTB 2L-VSCs with common DC-link

For larger tidal turbines with power ratings above 0.75 MW, multiple BTB VSC modules can be connected in parallel, accompanied by harmonic filters, as shown in Figures 12 and 13 [77]. In the case of Type 3 and Type 4 turbines, connecting VSC modules in parallel increases the current-carrying capacity [78]. This parallel configuration allows for power ratings of 1.5 MW for Type 4 turbines and up to 5 MW for Type 3 turbines [79]. A common DC-link across the converters helps reduce costs and space requirements [80].



Figure 12. Type 3 and 4 TCS with parallel connected BTB 2L-VSCs and common dc-link [81].



Figure 13. Type 3 and 4 TCS with parallel connected BTB 2L-VSCs, individual DC-links, and open winding transformer [81].

Self-commutated converters, which use pulse width modulation (PWM) and IGBTs, control both active and reactive power [82]. While PWM converters can meet reactive power demands, high-frequency switching generates harmonics, necessitating harmonic filters on both generator- and grid-side converters [83]. Generator-side harmonic filters reduce distortion in the generator's current, thus minimizing magnetic core and winding losses [84]. Grid-side filters ensure compliance with grid harmonic standards [85].

Finally, the grid-side harmonic filter output connects to a three-phase grid via a step-up transformer, switchgear, and circuit breaker [86]. In some cases, the converter can operate at collection point voltage levels, eliminating the need for a step-up transformer [87]. Efficient energy transfer from tidal stream turbines to the grid depends not only on converter technology but also on the grid connection layout. Properly designed grid connections ensure reliable transmission of energy over long distances while maintaining power quality and stability. The next section discusses grid connection layouts and their critical components, such as transformers, AC filters, and DC cables [88].

3.4. Definition of grid connection layout

In modern tidal stream energy systems, two primary transmission approaches are utilized for longdistance power transfer: a) AC collection-AC transmission and b) AC collection-DC transmission, as shown in Figure 13 [16]. The most used method is high-voltage alternating current (HVAC) transmission, depicted in Figure 14(a). HVAC is widely adopted due to its straightforward design and well-established technology. Typically, the generated voltage in tidal stream turbines is around 700 V, which is then stepped up to medium voltage levels such as 33 kV or, in specific cases, 66 kV, to facilitate transmission [17].

To handle over-voltage and charging current issues in subsea cables, shunt reactors are employed for power compensation. This ensures improved transmission capacity and efficiency, enabling smooth and reliable transfer of energy from offshore tidal systems to onshore grids [89].

3.4.1. Transformers and phase reactors

In tidal stream energy systems, a trio of single-phase transformers is employed to match the AC bus voltage with the input requirements of the power converter [89]. These transformers are equipped with tap changers for precise voltage control, enabling adjustments as needed. The design simplicity is attributed to the reduced harmonic content inherent in voltage source converter (VSC) high-voltage direct current (HVDC) systems [90]. This lower harmonic distortion eliminates the need for complex filtering solutions, leading to a more straightforward transformer configuration [91].



Figure 14. Standard (a) HVAC and (b) HVDC connections in offshore wind farms with elevated voltage levels [92].

3.4.2. AC filters

In tidal stream energy systems, managing high-order harmonics in AC output currents and voltages is critical [92]. These harmonics must be controlled to avoid interference with the AC system, as uncontrolled harmonics can cause equipment malfunctions or disrupt communication signals [89]. To achieve this, transmission systems use either passive or active high-pass filters. These filters help maintain nearly sinusoidal currents and voltages in the AC system. Thanks to their low harmonic

distortion and lack of need for reactive power compensation, these filters can be designed to be compact and simple [90].

3.4.3. DC cables and breakers

DC cables and breakers are essential components within the DC transmission system, working alongside DC links and earth electrodes. The development of DC cable technology has been driven by challenges in conventional cables, which are often heavy, difficult to install, and prone to faults [91]. Modern DC cables feature improved insulation, reduced power losses, lighter designs, and greater flexibility, making them suitable for applications such as submarine or overhead links [93]. In addition, hybrid DC breakers have emerged as a promising alternative to traditional mechanical or electronic switches. These hybrid breakers offer faster response times, enhanced efficiency, and greater reliability, with modular designs that allow easy adaptation to specific voltage and current demands [94].

3.4.4. DC capacitors and filters

DC capacitors and filters are critical in mitigating DC voltage ripple caused by the switching process in the converter's IGBTs [34]. To achieve this, the system uses two identical stacks of DC capacitors, each connected to a pole and grounded [90]. Additionally, DC filters can be applied on the transmission side to reduce interference with nearby metallic circuits. Thanks to the low harmonic content in the currents and voltages of VSC HVDC systems, these filters can be designed to be compact and straightforward [95].

3.4.4.1. HVDC system configurations

HVDC converters (Figure 15), which consist of rectifiers and inverters, provide flexibility in their interconnection through DC links, supporting various configurations [33]. These configurations include mono-polar HVDC systems, which can utilize ground return, metallic return, or a grounded midpoint [88]. Other configurations are back-to-back HVDC systems, bipolar HVDC systems (with variations such as bipole [95], bipole with metallic return, or bipole with series-connected converters), and multi-terminal HVDC systems, as shown in Figure 16 [96].



Figure 15. Different configurations of HVDC systems [97].



Figure 16. Typical planetary gearbox [26].

While power converters and grid connection layouts are vital for transmitting energy from tidal turbines to the grid, the mechanical transmission system is equally important for converting tidal flow into rotational energy for electricity generation. The following section focuses on mechanical transmission, including gearbox configurations and emerging trends in direct-drive systems.

3.5. Mechanical transmission

The gearbox is a crucial element of the mechanical transmission system. In large-scale wind turbines with significant rotor diameters, the rotor's rotational speed is typically low (approximately 18–50 rpm), which is insufficient for the efficient operation of generators that typically run between 1200 and 1800 rpm [45]. To address this mismatch, a gearbox is installed between the rotor's output shaft and the generator's input shaft, as shown in Figures 17 and 18, increasing the rotor speed to the required rpm for optimal power generation [47]. However, in systems utilizing multi-pole synchronous generators that are connected to the grid via a power converter, which electrically isolates the generator from the grid, the gearbox can be eliminated. This gearless configuration is particularly beneficial in offshore applications, where reduced maintenance is essential [49].

For wind turbines with smaller rotor diameters, the gearbox may also be unnecessary. A smaller rotor diameter results in a shorter distance traveled per revolution, potentially leading to a higher rotational speed that is compatible with the generator's requirements. This allows for a direct connection between the rotor and generator, similar to systems utilizing power converters. Smaller direct-drive wind turbines are commonly employed in standalone DC applications, such as battery charging [53].

The nacelle houses all of the turbine's key components, including the drive system (comprising the rotor shaft, gearbox, and bearings), the electrical generator, power electronics, yaw drive, mechanical brake, and control system, as shown in Figure 5. Since the nacelle must rotate to align with the wind direction, it is connected to the tower via bearings. The manufacturer's design choice to position the drive system and associated components above this rotational bearing is depicted in Figures 20–22 [98].



Figure 17. Geared tidal stream turbines: (a) Sea Gen tidal stream turbine (© Simec Atlantis Energy) [99].

According to the NREL gearbox reliability database, wind turbine gearbox reliability has improved substantially since 2013. Despite facing logistical challenges and a higher mean time to repair (MTTR) for offshore tidal turbines relative to onshore wind turbines, these advancements are significant [11]. Currently, gearbox failure rates are below 15%, with bearing failures representing 76% of these failures, while electrical system failures account for more than 25% [36]. The analysis highlights a notable decline in reliability for both standard and optimized gearboxes when compared to other industrial gearbox applications relevant to tidal stream turbines indicated in Figures 19 and 20. This decline is attributed to specific implementation factors; however, future gearbox designs remain unaffected [94]. This finding emphasizes the importance of both optimization and maintenance in improving gearbox reliability [100].



Figure 18. 650 kW planetary gearbox coupled with the generator of the MCT Sea Gen tidal stream turbine [101].



Figure 19. 1.5 MW planetary gearbox from the Mey Gen project [101].



Figure 20. 500 kW planetary gearbox employed in the TGL EMEC demonstrator project [102].

3.5.1. Addressing gearbox challenges

Addressing gearbox challenges in offshore tidal stream turbines necessitates adherence to rigorous operational and maintenance protocols. Effective lubrication is essential for optimal gearbox and transmission performance [6]. It mitigates friction, prevents excessive wear and heat generation, and preserves the integrity of critical components, such as gears, bearings, and seals. Lubrication issues can arise from contamination, degradation, leakage, or insufficient lubricant levels, potentially leading to increased noise, vibration, elevated temperatures, corrosion, and premature component failure [41]. To enhance gearbox reliability and longevity in challenging offshore environments, it is imperative to implement proactive measures to address these issues before they lead to failures [46]. Another prevalent maintenance challenge in tidal stream turbines involves alignment and balance issues. Proper alignment entails the precise positioning of shafts, gears, and bearings, while balance ensures the even distribution of mass and forces across rotating parts [57]. Misalignment and imbalance, often caused by improper installation, adjustment, or wear, can increase stress, vibration, noise, and accelerated component wear. These issues may also affect the efficiency and accuracy of power transmission [102].

Significant attention has been given to various configurations of permanent magnet generators for tidal stream turbines. The rim-driven concept is a notable option for direct-drive drivetrain systems, akin to the Open Hydro design (see Figure 21) [103]. In the rim-driven topology, the generator is positioned on the outer edge of the turbine, offering potential hydrodynamic efficiency advantages over a POD system, where the generator is housed within a nacelle [104]. Research has increasingly focused on various configurations of permanent magnet generators for tidal stream turbines. One promising approach is the rim-driven design, similar to the Open Hydro concept (see Figure 22) [105]. By positioning the generator on the turbine's outer edge, this design has the potential to improve efficiency compared to traditional nacelle-based systems [106].



Figure 21. The rim-driven concept using a radial flux permanent generator [6].



Figure 22. The concept of rim-drive utilizing a radial flux permanent generator [6].

4. Synchronous generators in tidal current energy converters (TCECs)

Marine current turbines can be classified in several ways, often with overlapping categories. This study focuses on categorizing these devices based on their interaction with water motion. The primary

classifications are:

- Horizontal-axis turbines
- Vertical-axis turbines
- Oscillating hydrofoil

To enhance the marine current velocity passing through the turbine and improve power capture, ducted structures are employed. These structures use the Venturi effect to channel flow toward the rotors, applicable to both horizontal and vertical turbines. This design approach increases the marine current speed in front of the rotor, potentially enhancing power output by up to 40% compared to turbines without ducts [6]. Additionally, it reduces turbulence and minimizes adverse effects on the rotor.

In tidal current energy converters (TCECs), turbines are equipped with two distinct types of synchronous generators: axial-flux and radial-flux direct-drive permanent magnet (DDPM) generators. The choice between these generators depends on the specific configuration requirements. Axial-flux generators are commonly used in pod-type structures, while radial-flux generators are employed in rim-driven configurations, as illustrated in Figure 23(a) [38]. For large-diameter radial-flux DDPM generators, a hollow center turbine structure, shown in Figure 23(b), is advantageous [106]. Radial-flux DDPM generators have traditionally been favored due to their development from induction generators [107]. However, axial-flux DDPM generators offer distinct benefits, such as increased compactness and power density, making them superior to conventional radial-flux DDPM generators in certain applications. Both axial and radial-flux DDPM generators, used in direct-drive systems, have garnered significant attention in both academic and industrial research [108].





Mechanical transmission systems deliver rotational energy to the generator, where it is converted into electrical power. The type and configuration of the generator plays a crucial role in determining the efficiency and reliability of tidal current energy converters (TCECs). The next section delves into synchronous generators, which are widely used in TCECs for their high efficiency and adaptability to variable-speed operations [108].

4.1. Single open-rotor horizontal-axis devices

A prevailing trend within major tidal energy device developers involves the utilization of a shared design foundation—the three-bladed, axial rotor concept indicated in Figure 24. This design approach, informed by the extensive operational expertise garnered from the wind industry, has become a prominent choice. Norway established a 300-kW grid-connected model in 2003. They have also commenced the development of their HS1000 1 MW turbine, scheduled for deployment by 2010. This innovative turbine is entirely submerged and follows a fixed yaw concept, meaning it remains stationary while its blades rotate with the tide's change. Consequently, the blades are required to have a symmetrical cross-section. The drivetrain is expected to incorporate a gearbox and a high-speed induction generator [109].



Figure 24. Single open-rotor horizontal-axis device (© Hammerfest Strøm AS) [109].

The turbine incorporates a robust system with multiple layers of redundancy, ensuring continuous operation even during faults or failures. It features an advanced condition monitoring system that predicts potential issues, enabling proactive maintenance [38]. Designed for a 25-year operational lifespan, the turbine includes three scheduled servicing intervals at 6¹/₄ years. With a total mass of 150 tons, it can be installed and retrieved using standard dynamically positioned (DP) construction vessels. Figure 25 illustrates this design, which integrates established wind sector principles with emerging tidal energy technologies [38].



Figure 25. The in-stream tidal turbine known as AR1500, which is anchored to the seabed, originates from SIMEC Atlantis [109].

The Sea Gen turbine, developed by Marine Current Turbines (MCT), was the world's first commercial-scale tidal turbine. Commissioned in July 2008 in Strangford Lough, Northern Ireland, it marked a significant advancement in tidal energy. The Sea Gen project, featuring two 600 kW turbines, had a total capacity of 1.2 MW and required a £12 million investment [38]. The model has been generating 5 GWh of power, meeting the annual energy needs of 1,500 households, as shown in Figure 26. Throughout its operational life, Sea Gen contributed 11.6 GWh to the grid before decommissioning in July 2019 [110]. Installed in May 2008, it was transported to Strangford Lough's mouth by barge, weighing 1000 tons with a 43 m wingspan. Designed by Peter Fraenkel, it operated like an *underwater windmill*, harnessing tidal currents to rotate its rotors. Built at Harland and Wolff shipyards in Belfast, Sea Gen was secured to the seabed in 14 days. It initially produced 150 kW of electricity during commissioning in July 2008, reaching full capacity by November 2008 [111].

With a mobile cross arm on a single 3 m diameter supporting pile, Sea Gen's twin rotors activated when tidal currents exceeded 1 m/s, achieving rotor tip speeds of approximately 12 m/s—approximately one-third of wind turbine speeds. The twin blades operated at a constant 14 rpm, driving a gearbox system [111].

The project, in partnership with ESB Independent Energy, supplied power to about 1,500 homes, with grid connection managed by Northern Ireland Electricity [20]. Deploying Sea Gen in the energetic tides of Strangford Lough presented unique technical challenges. As a pioneering underwater windmill, Sea Gen set a new precedent in tidal energy. Strangford Lough was a notable tidal hotspot in UK and Irish waters, alongside Anglesey, the Pentland Firth, and the Channel Islands [111].



Figure 26. Sea Gen [111].

In the summer of 2011, Atlantis Resources Corporation (now SAE Renewables) deployed its AR1000 tidal turbine at Berth 6 of the Fall of Warness tidal test site (see Figure 27). The AR1000 is a 1 MW horizontal-axis turbine, rated for a flow speed of 2.65 m/s, with an 18 m rotor diameter and a total weight of approximately 140 tons. The turbine was mounted on a 1,300-ton gravity-based structure and, when fully assembled, stood 22.5 m (73 feet) tall. Its control systems, including power conditioning and monitoring equipment, were housed in a dedicated control cabin on the island of Eday [21].

Following prototype testing at the European Marine Energy Centre (EMEC), the company advanced to the next-generation turbine, the AR1500. The AR1500 is being deployed as part of the Mey Gen project in the Pentland Firth. By March 2023, four 1.5 MW turbines had been installed at the Mey Gen site, which collectively generated 51 GWh of electrical energy for the national grid [111].



Figure 27. AR1000 tidal stream turbine (© Simec Atlantis Energy) [20].

Haiming I (10 kW) represents China's first long-term demonstration of a submarine vertical-axis tidal current power generation system (TCPGS), as illustrated in Figure 28. This power station, measuring 9.0 m \times 7.5 m \times 6.5 m and weighing 20 tons, is equipped with a horizontal-axis fixed-pitch turbine, an expansion-type deflector, and a direct-drive permanent magnet synchronous generator (PMSG). The station is supported by a three-legged base frame that can be raised for maintenance. Additionally, the platform incorporates a 180° self-adjusting reversing mechanism, allowing it to efficiently accommodate bidirectional currents [112].



Figure 28. Haiming I: 10 kW submarine vertical-axis TCPG [112].

Hai Neng II (2×100 kW), also known as the Zhaitang Island Isolated Hybrid Power Demonstration Station, is equipped with a cruciform carrier, two columns, and four flexible mooring systems, as illustrated in Figure 29. Each column supports a horizontal-axis variable-pitch turbine with a 12 m diameter [10]. These turbines are directly coupled to low-speed permanent magnet synchronous generators (PMSGs), each with a capacity of 100 kW, and are designed to operate in bidirectional tidal currents. The turbines have a rated flow velocity of 1.7 m/s, with a cut-in speed of 0.6 m/s and a cut-out speed of 2 m/s [11].



Figure 29. "Hai Neng II" 2 × 100 kW floating horizontal-axis TCPGS [11].

NTU and NTOU's collaboration led to the development of the Floating Kuroshio Turbine (FKT) system. Laboratory tests of a 1/25 scale model demonstrated a power generation capability of 850 W at a towing speed of 1.45 m/s. A feasibility study for a 1/5 scale model is underway, with expected power outputs of up to 20 kW [113].

The NSYSU team introduced the floating nozzle-diffusor duct (NDD) power generation system, depicted in Figure 30. Deployed near the Penghu cross-sea bridge in 2013, this system achieved a power take-off (PTO) of 5 kW during on-site testing. A new design iteration with a circular nozzle-diffusor duct is projected to triple the PTO compared to the previous model [113]. Additionally, a collaboration between NSYSU, Tainan Hydraulics Laboratory (THL), and Wanchi Steel Industry Company developed a 50 kW Kuroshio Ocean current power harvester. During a 60 h open-water towing test conducted in southeast Taiwan in 2016, the system achieved an average output of 26 kW at a current speed of 1.27 m/s [113].



Figure 30. Prototype of Floating Kuroshio Turbine (FKT) [113].

One such innovation involves the incorporation of ducted horizontal-axis devices, which enhance energy capture and resilience through refined flow dynamics. The following section delves into the design and performance benefits of these ducted systems, offering a new perspective on optimizing tidal energy extraction.

4.2. Ducted horizontal-axis devices in tidal turbines: enhancing efficiency and performance

The use of tidal currents presents unique design opportunities for turbines, particularly with ducted horizontal axis devices (see Figure 31). Unlike wind turbines, tidal turbines benefit from smaller rotor diameters, which facilitate the integration of ducting mechanisms to enhance flow dynamics. This innovation improves energy capture per unit of rotor area [114].

Ducted systems offer several advantages for tidal turbines. They enhance resilience to off-axis flow disturbances, a challenge often encountered by wind turbines. Additionally, supporting turbine blades at their extremities is feasible, which enhances load resistance and structural integrity. Key to the design of these turbines is their adaptability to tidal flow variations. Turbines must operate efficiently in both directions to accommodate changing tidal flows. Incorporating bi-directional rotors is essential for this purpose. As depicted in Figure 32, this approach redefines energy extraction potential from tidal currents, increasing efficiency, resilience to flow fluctuations, and performance under varying tidal conditions [115].



Figure 31. Illustration of ducted horizontal-axis tidal turbines (Open Hydro) [28].

The next section explores multiple open-rotor horizontal-axis devices, highlighting their potential to enhance energy generation capacity and operational efficiency in tidal stream systems.

4.3. Multiple open-rotor horizontal axis devices

Various rotor concepts aim to maximize generating capacity on each foundation, overcoming limitations imposed by single rotor sizes. These concepts address issues such as asymmetric loading, dynamic behavior during failures, and maintenance considerations, as illustrated in Figure 32.





4.4. Transverse axis devices

The Darrieus-type wind turbine and vertical-axis marine current turbines function as lift-based systems with the generation axis perpendicular to the flow [114]. While they may have lower theoretical efficiency compared to horizontal-axis machines, they offer benefits in structural efficiency and maintenance. Some designs are exploring ducting to enhance energy capture. A significant concept is the modular tidal fence array, where multiple devices are interconnected. Economic considerations include the size of the required ducting, as shown in Figure 33 [114].



Figure 33. Transverse axis devices [114].

Table 1 summarizes experimental twin-rotor tidal current turbines worldwide. Most of these turbines are in the demonstration phase and have not yet reached commercial deployment [116]. The floating catamaran, among various floating mooring platforms, is particularly promising. It provides a spacious deck, facilitates maintenance and deployment, and adapts well to varying water depths and complex seabed topographies [117].

Institutions	Location	Turbine name	Turbine number	Platform	Total capacity	Operation year
MCT-Atlantis Resources	Kyle Rhea (Scotland, UK)	Sea Gen	2	Pile foundation	2 × 600 W	≥2016
AK-1000	Singapore	AK-1000	2	Seabed mounted	$2 \times 500 \text{ W}$	≥2010
SMD Hydro Vision Ltd	Orkney, UK	Tidel	2	Floating foundation	$2 \times 500 \text{ W}$	≥2006
National Taiwan Ocean University	Coast of Taiwan	Kurshio	2	Floating foundation	2 × 10 W	≥2016
Harbin Engineering University	Zhejiang, China	Hai Neng I	2	Floating foundation (catamaran)	2 × 150 W	≥2010
Harbin Engineering University	Qingdao, China	Hai Neng II	2	Floating foundation (catamaran)	2 × 100 W	2013
Harbin Engineering University	Zhejiang, China	Hai Neng III		Floating foundation (catamaran)	2 × 300 W	≥2013

Table 1. Pilot floating two-rotor tidal current turbines in the world [11].

While the development of individual tidal energy devices continues to evolve, hybrid configurations that integrate multiple renewable energy sources present a compelling solution for overcoming the limitations of standalone systems. By combining wind and tidal stream technologies, hybrid systems capitalize on the complementary characteristics of these resources, ensuring greater reliability and efficiency in electricity generation. The following section examines the growing trend toward multi-energy systems, emphasizing the benefits of integrating wind and tidal energy technologies into a unified framework.

5. Hybrid configurations integrating wind and tidal stream technologies

Exclusive reliance on wind energy for electricity generation is significantly affected by environmental variability. By integrating multiple renewable sources—such as wind and tidal, each with unique characteristics and advantages—it is possible to mitigate individual limitations and enhance overall system efficiency. Recent developments reflect a growing trend toward multi-energy complementary systems.

5.1. Hybrid system combining offshore wind and tidal turbines

Hybrid energy models that combine tidal and wind power technologies are increasingly explored. These systems typically feature separate subsystems for tidal and wind energy, sharing a common platform [118]. Combining wind and tidal energy can stabilize power output and reduce fluctuations. However, optimizing multi-energy systems requires advanced assessment technologies to account for the varying characteristics of different energy sources [119].

5.2. Offshore wind and tidal concepts

Hybrid systems that merge wind and tidal technologies are still under development. Two notable designs include:

- Savonius Keel and Wind Turbine Darrieus (SKWID), shown in Figure 34, is a unique prototype developed by Mitsui Ocean Development and Engineering Company (MODEC). This system combines a Savonius current turbine with a Darrieus wind turbine and is designed for deployment off the coast of Japan. Its primary goal is to provide reliable backup power to isolated islands, offering a sustainable energy solution for remote locations [120].
- Marine Current Turbines' Patented Device: Inspired by the Sea Gen tidal apparatus, this concept features multiple tidal turbines mounted on a single-column structure. Though still in the conceptual phase, it proposes elevating turbines above water to improve efficiency [120].

Combining the outputs of tidal and wind turbines results in a total annual energy generation of 96.41 GWh, which is double that of using tidal turbines alone [121].



Figure 34. (a) MODEC's SKWID, Savonius Keel and Wind Turbine Darrieus. (b) Marine Current Turbine's patented turbine that combines air and water flow [119].

The Hybrid Offshore Wind and Tidal Turbine (HOTT) system integrates tidal current generation to manage wind power fluctuations, as shown in Figure 35 [120]. This hybrid approach aims to stabilize power production and enhance system reliability, particularly for remote islands where extending the electricity grid is economically unfeasible [121]. This approach offers a viable solution for sustainable energy collection, with future research focusing on storage, grid integration, power regulation, cost optimization, and environmental impact assessment [122].

Various connection methods are available, with the DC bus-based approach selected for its cost-effectiveness and simplified control [123]. The focus remains on capital expenditures (CAPEX) and operational expenditures (OPEX), which are critical for optimizing energy production costs. Addressing potential instability from renewable energy integration will be essential as cumulative renewable capacity increases [124].



Figure 35. Illustration depicting the concept of a hybrid offshore wind-tidal (HOTT) system [125].

6. Conclusions

The integration of offshore wind and tidal stream technologies offers a transformative approach to addressing the increasing energy demands of coastal regions. This hybrid approach provides a sustainable pathway toward decarbonization, with tidal energy emerging as a vital component due to its predictability and advancing technology. Although challenges such as cost-effectiveness, scalability, and reliability persist, this study highlights the significant potential of hybrid systems, particularly those leveraging resources like the Agulhas Current, to enhance energy stability, reliability, and efficiency.

Key developmental pathways identified in this study include advancing large-scale turbines for grid integration for localized marine applications. To further the practical adoption of these systems, future research must address the technical and economic challenges in a focused and systematic manner.

Future research directions

To guide continued progress, the following specific areas are proposed for future exploration:

1. Advanced materials and structural designs

- Investigate innovative materials, including corrosion-resistant alloys and composites, to enhance durability and reduce operational costs in harsh marine environments.
- Explore biomimetic designs inspired by aquatic organisms to improve turbine efficiency and minimize ecological disruption.
- 2. Computational methodologies and AI integration
- Develop AI-driven optimization tools to enhance turbine performance, predictive maintenance, and control strategies for hybrid wind-tidal systems.
- Incorporate machine learning in energy management systems and power converter design to improve grid integration and real-time adaptability.

3. **Energy storage solutions**

- Focus on hybrid storage systems combining short-term (e.g., super-capacitors) and long-term (e.g., advanced battery technologies) solutions to stabilize power supply.
- Explore modular and scalable storage designs tailored to the unique intermittency profiles of wind and tidal energy.
- 4. **Power conversion and grid integration**
- Develop advanced converter technologies to improve power quality, reduce harmonic distortion, and ensure seamless integration into national grids.
- Investigate HVDC and advanced AC transmission systems for efficient energy transport from offshore systems to onshore grids.
- 5. Environmental impact and ecosystem integration
- Conduct detailed impact assessments to ensure minimal disruption to marine ecosystems, with a focus on species migration patterns and habitat preservation.
- Implement AI- and IoT-enabled adaptive monitoring systems for real-time assessment and mitigation of ecological effects.

Broader prospects

Looking ahead, scaling pilot projects into full-scale commercial installations will be critical,

particularly in regions with high energy potential, such as the Agulhas Current. Success in this domain will require robust interdisciplinary collaboration among engineers, environmental scientists, economists, and policymakers. Additionally, fostering international partnerships and sharing best practices from successful projects can accelerate technological innovation and cost reduction in offshore hybrid energy systems. By addressing these challenges and seizing growth opportunities, hybrid wind-tidal systems can play a pivotal role in achieving global sustainability and energy resilience goals.

Use of AI tools declaration

The authors declare that no Artificial Intelligence (AI) tools were used in the creation of this article, including in the drafting, editing, or data analysis processes.

Acknowledgments

Acknowledge the CPUT bursary award.

Conflict of interest

The authors declare that they have no conflicts of interest regarding the publication of this paper.

Author contributions

Ladislas Kangaji led the research, conducted the literature review, and was responsible for the collection and organization of review papers. He also drafted the manuscript and coordinated the writing process. Atanda Raji provided critical insights, guided the development of the research methodology, and played a key role in reviewing and revising the manuscript. Efe Orumwense contributed to the conceptual framework of the study and provided valuable feedback during the manuscript editing process.

References

- 1. Rautenbach C, Barnes MA, de Vos M (2019) Tidal characteristics of South Africa. *Deep Sea Res Part I: Oceanogr Res Pap* 150: 103079. https://doi.org/10.1016/j.dsr.2019.103079
- 2. Verma J, Kumar D (2021) Recent developments in energy storage systems for marine environment. *Mater Adv* 2: 6800–6815. https://doi.org/10.1039/D1MA00746G
- 3. Nasab NM, Kilby J, Bakhtiaryfard L (2022) Integration of wind and tidal turbines using spar buoy floating foundations. 10: 1165–1189. https://doi.org/10.3934/energy.2022055
- 4. Melikoglu M (2017) Current status and future of ocean energy sources : A global review. *Ocean Eng* 148: 563–573. https://doi.org/10.1016/j.oceaneng.2017.11.045
- Khan N, Kalair A, Abas N, et al. (2017) Review of ocean tidal, wave and thermal energy technologies. *Renewable Sustainable Energy Rev* 72: 590–604. https://doi.org/10.1016/j.rser.2017.01.079

- Rehman S, Alhems LM, Alam M, et al. (2022) A review of energy extraction from wind and ocean : Technologies, merits, efficiencies, and cost. *Ocean Eng* 267: 113192. https://doi.org/10.1016/j.oceaneng.2022.113192
- Mtukushe NF, Ojo EE (2021) The study of electrical power generation from tidal energy in South Africa. 2021 South. African Univ. Power Eng. Conf. Mechatronics/Pattern Recognit. Assoc. South Africa, SAUPEC/RobMech/PRASA 2021, 7–12. https://doi.org/10.1109/SAUPEC/RobMech/PRASA52254.2021.9377011
- Akinbami OM, Oke SR, Bodunrin MO (2021) The state of renewable energy development in South Africa: An overview. *Alexandria Eng J* 60: 5077–5093. https://doi.org/10.1016/j.aej.2021.03.065
- 9. Tarafdar S, Abroshan M, Mirsalim M, et al. (2013) Permanent magnet linear synchronous generator for an oscillating hydrofoil in a tidal current regime. *Int J Eng Technol* 5: 231–236. https://doi.org/10.7763/IJET.2013.V5.549
- Khooban MH, Gheisarnejad M (2020) Islanded microgrid frequency regulations concerning the integration of tidal power units: Real-Time implementation. *IEEE Trans Circuits Syst II Express Briefs* 67: 1099–1103. https://doi.org/10.1109/TCSII.2019.2928838
- 11. Ranjan T, Thanthirige M, Goggins J, et al. (2023) A state-of-the-art review of structural testing of tidal turbine blades. *Energies* 16: 4061. https://doi.org/10.3390/en16104061
- 12. Funke SW, Farrell PE, Piggott MD (2014) Tidal turbine array optimisation using the adjoint approach. *Renewable Energy* 63: 658–673. https://doi.org/10.1016/j.renene.2013.09.031
- Zeyringer M, Fais B, Keppo I, et al. (2018) The potential of marine energy technologies in the UK—Evaluation from a systems perspective. *Renewable Energy* 115: 1281–1293. https://doi.org/10.1016/j.renene.2017.07.092
- 14. Fard RN, Tedeschi E (2018) Integration of distributed energy resources into offshore and subsea grids. 3: 36–45. https://doi.org/10.24295/CPSSTPEA.2018.00004
- 15. Rahman ML, Oka S, Shirai Y (2010) Hybrid power generation system using offshore-wind turbine and tidal turbine for power fluctuation compensation (HOT-PC). *IEEE Trans Sustainable Energy* 1: 92–98. https://doi.org/10.1109/TSTE.2010.2050347
- Raza SA, Ali SW, Sadiq M, et al. (2021) Offshore wind farm-grid integration: A review on infrastructure, challenges, and grid solutions. *IEEE Access* 9: 102811–102827. https://doi.org/10.1109/ACCESS.2021.3098705
- 17. Boersma S, Doekemeijer BM, Gebraad PMO, et al. (2017) A tutorial on control-oriented modeling and control of wind farms. 2017 American Control Conference (ACC), 1–18. https://doi.org/10.23919/ACC.2017.7962923
- Soukissian TH, Denaxa D, Karathanasi F, et al. (2017) Marine renewable energy in the Mediterranean Sea: Status and perspectives. *Energies* 10: 1–56. https://doi.org/10.3390/en10101512
- 19. Li J, Wang G, Li Z, et al. (2020) A review on development of offshore wind energy conversion system. *Int J Energy Res* 44: 9283–9297. https://doi.org/10.1002/er.5751
- 20. Tavakoli S, Khojasteh D, Haghani M, et al. (2023) A review on the progress and research directions of ocean engineering. Ocean Eng 272: 113617. https://doi.org/10.1016/j.oceaneng.2023.113617
- 21. Zhou Y (2022) Ocean energy applications for coastal communities with artificial intelligence— A state-of-the-art review. *Energy AI* 10: 100189. https://doi.org/10.1016/j.egyai.2022.100189

- 22. Shao H, Henriques R, Morais H, et al. (2023) Power quality monitoring in electric grid integrating offshore wind energy: A review. *Renewable Sustainable Energy Rev* 191: 114094. https://doi.org/10.1016/j.rser.2023.114094
- Kai LY, Sarip S, Kaidi HM, et al. (2021) Current status and possible future applications of marine current energy devices in Malaysia: A review. *IEEE Access* 9: 86869–86888. https://doi.org/10.1109/ACCESS.2021.3088761
- Itiki R, Di Santo SG, Itiki C, et al. (2019) A comprehensive review and proposed architecture for offshore power system. *Int J Electr Power Energy Syst* 111: 79–92. https://doi.org/10.1016/j.ijepes.2019.04.008
- 25. Petracca E, Faraggiana E, Ghigo A, et al. (2022) Design and techno-economic analysis of a novel hybrid offshore wind and wave energy system. *Energies* 15: 2739. https://doi.org/10.3390/en15082739
- Zhang J, Moreau L, Machmoum M, et al. (2014) State of the art in tidal current energy extracting technologies. 2014 First International Conference on Green Energy ICGE 2014, 1–7. https://doi.org/10.1109/ICGE.2014.6835388
- 27. Strasser T, Andrén F, Kathan J, et al. (2015) A review of architectures and concepts for intelligence in future electric energy systems. *IEEE Trans Ind Electron* 62: 2424–2438. https://doi.org/10.1109/TIE.2014.2361486
- 28. Zainol MZ, Ismail N, Zainol I, et al. (2017) A review on the status of tidal energy technology worldwide. Sci Int 29: 659–667. Available from: https://www.researchgate.net/publication/317617074.
- Kalair A, Abas N, Kalair AR, et al. (2017) Review of harmonic analysis, modeling and mitigation techniques. *Renewable Sustainable Energy Rev* 78: 1152–1187. https://doi.org/10.1016/j.rser.2017.04.121
- Wei S, Zhang L, Xu Y, et al. (2017) Hierarchical optimization for the double-sided ring structure of the collector system planning of large offshore wind farms. *IEEE Trans Sustainable Energy* 8: 1029–1040. https://doi.org/10.1109/TSTE.2016.2646061
- 31. Pape M, Kazerani M (2022) A generic power converter sizing framework for series-connected DC offshore wind farms. *IEEE Trans Power Electron* 37: 2307–2320. https://doi.org/10.1109/TPEL.2021.3106578
- 32. Bose BK (2017) Artificial intelligence techniques in smart grid and renewable energy systems some example applications. *Proc IEEE* 105: 2262–2273. https://doi.org/10.1109/JPROC.2017.2756596
- 33. Robins PE, Neill SP, Lewis MJ, et al. (2015) Characterising the spatial and temporal variability of the tidal-stream energy resource over the northwest European shelf seas. *Appl Energy* 147: 510–522. https://doi.org/10.1016/j.apenergy.2015.03.045
- 34. Carlos EWUL (2014) Performance of pitch and stall regulated tidal stream turbines. *IEEE Trans Sustainable Energy* 5: 64–72. https://doi.org/10.1109/TSTE.2013.2272653
- 35. Kaldellis JK, Boulogiorgou D (2024) Renewable energy: Wind energy. *Living With Climate Change* 2024: 513–557. https://doi.org/10.1016/B978-0-443-18515-1.00017-4
- 36. Rafiei M, Salvatore F, Capponi FG (2019) Generator topologies for horizontal axis tidal turbine. *Lect Notes Electr Eng* 615: 447–459. https://doi.org/10.1007/978-3-030-37161-6_34

- Li G, Zhu W (2023) Tidal current energy harvesting technologies : A review of current status and life cycle assessment. *Renewable Sustainable Energy Rev* 179: 113269. https://doi.org/10.1016/j.rser.2023.113269
- 38. King J, Tryfonas T (2009) Tidal stream power technology—state of the art. *OCEANS 2009-EUROPE*. https://doi.org/10.1109/OCEANSE.2009.5278329
- 39. Khojasteh D, Shamsipour A, Huang L, et al. (2023) A large-scale review of wave and tidal energy research over the last 20 years. Ocean Eng 282: 114995. https://doi.org/10.1016/j.oceaneng.2023.114995
- Abd Rahim MW, Rahman AA, Izham M, et al. (2023) Tidal energy in Malaysia: An overview of potentials, device suitability, issues and outlook. *Reg Stud Mar Sci* 61: 102853. https://doi.org/10.1016/j.rsma.2023.102853
- 41. Gattuso J, Magnan AK, Bopp L, et al. (2018) Ocean solutions to address climate change and its effects on marine ecosystems. https://doi.org/10.3389/fmars.2018.00337
- 42. Touimi K, Benbouzid M (2020) Optimal Design of a multigrid permanent magnet. 13: 487. https://doi.org/10.3390/en13020487
- 43. Sousounis MC (2018) Electro-Mechanical Modelling of Tidal Arrays. *The University of Edinburgh*, Ph.D. dissertation. Available from: http://hdl.handle.net/1842/31089.
- 44. Zhang Y, Zhao Y, Sun W, et al. (2021) Ocean wave energy converters : Technical principle, device realization, and performance evaluation. *Renewable Sustainable Energy Rev* 141: 110764. https://doi.org/10.1016/j.rser.2021.110764
- 45. Gustavo M, Gimenez J (2011) Technical and regulatory exigencies for grid connection of wind generation. *Wind Farm—Tech Regul Potential Estim Siting Assess*. https://doi.org/10.5772/16474
- 46. Blaabjerg F, Liu H, Loh PC (2014) Marine energy generation systems and related monitoring and control. *IEEE Instrum Meas Mag* 17: 27–32. https://doi.org/10.1109/MIM.2014.6810042
- 47. Alcorn R, O'Sullivan DL (2013) Electrical design for ocean energy wave and tidal energy systems. *Inst Eng Technol*, 53. https://doi.org/10.13140/2.1.1934.8804
- 48. Marten A, Akmatov V, Sørensen TB, et al. (2018) Kriegers flak-combined grid solution: coordinated cross-border control of a meshed HVAC/HVDC offshore wind power grid. https://doi.org/10.1049/iet-rpg.2017.0792
- 49. Sousounis MC, Shek JKH (2017) Mitigation of torque pulsations in variable pitch tidal current turbines using speed control. *Proc 12th Eur Wave Tidal Energy Conf.* https://doi.org/10.13140/RG.2.2.32130.17603
- 50. Baykov A, Dar'enkov A, Kurkin A, et al. (2019) Mathematical modelling of a tidal power station with diesel and wind units. *J King Saud Univ—Sci* 31: 1491–1498. https://doi.org/10.1016/j.jksus.2019.01.009
- 51. Wang L, Lin CY, Wu HY, et al. (2018) Stability analysis of a microgrid system with a hybrid offshore wind and ocean energy farm fed to a power grid through an HVDC link. *IEEE Trans Ind Appl* 54: 2012–2022. https://doi.org/10.1109/TIA.2017.2787126
- 52. Karimirad M (2014) *Offshore Energy Structures*, 1st edition, Cham, Switzerland: Springer Cham Heidelberg New York Dordrecht London. https://doi.org/10.1007/978-3-319-12175-8
- 53. Rusu E (2013) The expected dynamics of the european offshore wind sector in the climate change context. *J Mar Sci Eng* 11: 1967. https://doi.org/10.3390/jmse11101967

- 54. Alfahi STY, Alkahtani AA, Al-Shetwi AQ, et al. (2021) Supraharmonics in power grid: identification, standards, and measurement techniques. *IEEE Access* 9: 103677–103690. https://doi.org/10.1109/ACCESS.2021.3099013
- 55. Errami Y, Ieee SM, Maaroufi M (2011) Modelling and control strategy of PMSG based variable speed wind energy conversion system. 2011 Int Conf Multimed Comput Syst 2: 1–6. https://doi.org/10.1109/ICMCS.2011.5945736
- 56. Mousavi GSM (2012) An autonomous hybrid energy system of wind/tidal/microturbine/battery storage. Int J Electr Power Energy Syst 43: 1144–1154. https://doi.org/10.1016/j.ijepes.2012.05.060
- 57. Cheng M, Zhu Y (2014) The state of the art of wind energy conversion systems and technologies : A review. *Energy Convers Manag* 88: 332–347. https://doi.org/10.1016/j.enconman.2014.08.037
- Liu M, Sun Z, Liu G, et al. (2019) Study on the influence of large-scale wind power integration on transient stability of power system. APAP 2019—8th IEEE Int Conf Adv Power Syst Autom Prot, 1156–1159. https://doi.org/10.1109/APAP47170.2019.9224650
- 59. Abo-Khalil AG, Alghamdi AS (2021) MPPT of permanent magnet synchronous generator in tidal energy systems using support vector regression. *Sustainability* 13: 1–15. https://doi.org/10.3390/su13042223
- Potgieter JHJ, Kamper MJ (2015) Design Optimization of directly grid-connected PM machines for wind energy applications. *IEEE Trans Ind Appl* 51: 2949–2958. https://doi.org/10.1109/TIA.2015.2394506
- 61. Krylov D, Kholod O, Radohuz S (2020) Active rectifier with different control system types. 2020 *IEEE 4th International Conference on Intelligent Energy and Power Systems (IEPS)*, Kyiv, Ukraine. https://doi.org/10.1109/IEPS51250.2020.9263226
- 62. Yan J, Lin H, Feng Y, et al. (2014) Control of a grid-connected direct-drive wind energy conversion system. 66: 371–380. https://doi.org/10.1016/j.renene.2013.12.037
- 63. Krylov DS, Kholod O (2021) The efficiency of the active-controlled rectifier operation in the mains voltage distortion mode. 2: 30–35. https://doi.org/10.20998/2074-272X.2021.2.05
- 64. Chhipa AA, Chakrabarti P, Bolshev V, et al. (2022) Modeling and control strategy of wind energy conversion system with grid-connected doubly-fed induction generator. *Energies*, 15. https://doi.org/10.3390/en15186694
- 65. Tripathi RN, Singh A, Hanamoto T (2015) Design and control of LCL filter interfaced grid connected solar photovoltaic (SPV) system using power balance theory. *Int J Electr Power Energy Syst* 69: 264–272. https://doi.org/10.1016/j.ijepes.2015.01.018
- 66. García-Triviño P, Gil-Mena AJ, Llorens-Iborra F, et al. (2015) Power control based on particle swarm optimization of grid-connected inverter for hybrid renewable energy system. *Energy Convers Manag* 91: 83–92. https://doi.org/10.1016/j.enconman.2014.11.051
- 67. Vukobratović M, Marić P, Nikolovski S, et al. (2018) Distributed generation harmonic interaction in the active distribution network. *Teh Vjesn* 25: 1720–1730. https://doi.org/10.17559/TV-20171025123650
- Sousounis MC, Shek JKH, Mueller MA (2016) Modelling, control and frequency domain analysis of a tidal current conversion system with onshore converters. *IET Renewable Power Gener* 10: 158–165. https://doi.org/10.1049/iet-rpg.2014.0331

- Moreira AB, Dos Santos Barros TA, De Castro Teixeira VS, et al. (2019) Control of powers for wind power generation and grid current harmonics filtering from doubly fed induction generator: comparison of two strategies. *IEEE Access* 7: 32703–32713. https://doi.org/10.1109/ACCESS.2019.2899456
- 70. Naderipour A, Zin AAM, Bin Habibuddin MH, et al. (2017) An improved synchronous reference frame current control strategy for a photovoltaic grid-connected inverter under unbalanced and nonlinear load conditions. *PLoS One* 12: 1–17. https://doi.org/10.1371/journal.pone.0164856
- Patil A, Gadgune S (2022) DC side voltage regulation of three phase PWM rectifier control using sliding mode controller. 2022 International Conference on Futuristic Technologies, INCOFT 2022, 1–4. https://doi.org/10.1109/INCOFT55651.2022.10094557
- 72. Abu-Siada A, Budiri J, Abdou AF (2018) Solid state transformers topologies, controllers, and applications: State-of-the-art literature review. *Electronics*, 7. https://doi.org/10.3390/electronics7110298
- Reznik A, Simões MG, Al-Durra A, et al. (2012) LCL filter design and performance analysis for small wind turbine systems. *PEMWA 2012—2012 IEEE Power Electron Mach Wind Appl*, 1–7. https://doi.org/10.1109/PEMWA.2012.6316408
- 74. Cai Y, Member S, He Y, et al. (2023) Integrated design of filter and controller parameters inverter based on harmonic state-space model. *IEEE Trans Power Electron* 38: 6455–6473. https://doi.org/10.1109/TPEL.2023.3241091
- 75. Ranjan A, Giribabu D (2023) Design of LCL filter for grid connected three phase three level inverter to meet IEEE 519 standards. *2023 IEEE Int Students' Conf Electr Electron Comput Sci SCEECS*, 1–6. https://doi.org/10.1109/SCEECS57921.2023.10061817
- 76. Zhan P, Lin W, Wen J, et al. (2012) Design of LCL filters for the back-to-back converter in a doubly fed induction generator. 2012 IEEE Innov Smart Grid Technol—Asia, ISGT Asia 2012, 1– 6. https://doi.org/10.1109/ISGT-Asia.2012.6303305
- 77. Reznik A, Simoes MG, Al-Durra A, et al. (2014) LCL filter design and performance analysis for grid-interconnected systems. *IEEE Trans Ind Appl* 50: 1225–1232. https://doi.org/10.1109/TIA.2013.2274612
- 78. Musasa K, Nwulu NI, Gitau MN, et al. (2017) Review on DC collection grids for offshore wind farms with high-voltage DC transmission system. *IET Power Electron* 10: 2104–2115. https://doi.org/10.1049/iet-pel.2017.0182
- 79. Mansouri A, El A, Lajouad R et al. (2023) Wind energy based conversion topologies and maximum power point tracking: A Comprehensive Review and Analysis. *e-Prime—Adv Electr Eng Electron Energy* 6: 100351. https://doi.org/10.1016/j.prime.2023.100351
- 80. Liang X (2017) Emerging power quality challenges due to integration of renewable energy sources. *IEEE Trans Ind Appl* 53: 855–866. https://doi.org/10.1109/TIA.2016.2626253
- Yaramasu V, Wu B, Sen PC, et al. (2015) High-power wind energy conversion systems: State-ofthe-art and emerging technologies. *Proc IEEE* 103: 740–788. https://doi.org/10.1109/JPROC.2014.2378692
- Sanjuan SL (2010) Voltage oriented control of three-phase boost PWM converters: design, simulation, and implementation of a 3-phase boost battery charger. Chalmers University of Technology, Gothenburg, Sweden, M.S. thesis. Available from: https://doi:publications.lib.chalmers.se/records/fulltext/173977/173977.

- Ni K, Hu Y, Liu Y, et al. (2017) Performance analysis of a four-switch three-phase grid-side converter with modulation simplification in a doubly-fed induction generator-based wind turbine (DFIG-WT) with different external disturbances. *Energies*, 10. https://doi.org/10.3390/en10050706
- Amenedo JLR, Gomez SA, Alonso-Martinez J, et al. (2021) Grid-Forming converters control based on the reactive power synchronization method for renewable power plants. *IEEE Access* 9: 67989–68007. https://doi.org/10.1109/ACCESS.2021.3078078
- Tola OJ, Tsado J, Umoh EA, et al. (2022) Modeling and simulation of marine current energy conversion system with six-phase permanent magnet synchronous generator. *Eurasia Proc Sci Technol Eng Math* 21: 1–10. https://doi.org/10.55549/epstem.1224038
- 86. Brantsæter H, Kocewiak Ł, Årdal AR, et al. (2015) Passive filter design and offshore wind turbine modelling for system level harmonic studies. *Energy Procedia* 80: 401–410. https://doi.org/10.1016/j.egypro.2015.11.444
- Deng J, Cheng F, Yao L, et al. (2023) A review of system topologies, key operation and control technologies for offshore wind power transmission based on HVDC. *IET Gener Transm Distrib* 17: 3345–3363. https://doi.org/10.1049/gtd2.12894
- Jain A, Shankar S, Vanitha V (2018) Power generation using Permanent Magnet Synchronous Generator (PMSG) based variable speed wind energy conversion system (WECS): An overview. *J Green Eng* 7: 477–504. https://doi.org/10.13052/jge1904-4720.742
- Korompili A, Wu Q, Zhao H (2016) Review of VSC HVDC connection for offshore wind power integration. *Renewable Sustainable Energy Rev* 59: 1405–1414. https://doi.org/10.1016/j.rser.2016.01.064
- 90. Patra RR, Asha RMA, Krishna PB (2023) Active power control of Hybrid-HVDC line connected to offshore DFIG-based wind farm during mainland grid voltage fluctuations. 2023 IEEE IAS Glob Conf Renew Energy Hydrog Technol GlobConHT 2023, 1–7. https://doi.org/10.1109/GlobConHT56829.2023.10087847
- 91. Roy A, Auger F, Dupriez-Robin F, et al. (2018) Electrical power supply of remote maritime areas: A review of hybrid systems based on marine renewable energies. *Energies*, 11. https://doi.org/10.3390/en11071904
- 92. Krneta N, Sano K (2024) AC fault ride-through method of HVDC interconnector for islanded AC grid with offshore wind farm and loads. *IEEE Access* 12: 62099—62106. https://doi.org/10.1109/ACCESS.2024.3394322
- Sousounis MC, Shek JKH, Mueller MA (2016) Filter design for cable overvoltage and power loss minimization in a tidal energy system with onshore converters. *IEEE Trans Sustainable Energy* 7: 400–408. https://doi.org/10.1109/TSTE.2015.2424258
- 94. Chen H, Tang T, Ait-Ahmed N, et al. (2018) Attraction, challenge and current status of marine current energy. *IEEE Access* 6: 12665–12685. https://doi.org/10.1109/ACCESS.2018.2795708
- 95. Abdelsalam I, Adam GP, Williams BW (2016) Current source back-to-back converter for wind energy conversion systems. *IET Renewable Power Gener* 10: 1552–1561. https://doi.org/10.1049/iet-rpg.2016.0177
- 96. Rajendran S, Diaz M, Cárdenas R, et al. (2022) A review of generators and power converters for multi-mw wind energy conversion systems. *Processes*, 10. https://doi.org/10.3390/pr10112302

- 97. Raza M (2021) Voltage and reactive power correlation in multi-objective optimization of an offshore grid. *Automatika* 62: 454–470. https://doi.org/10.1080/00051144.2021.1984632
- 98. Garcia P, Kyozuka Y (2021) Tidal stream energy as a potential continuous power producer: A case study for West Japan. *Energy Convers Manag* 245: 114533. https://doi.org/10.1016/j.enconman.2021.114533
- 99. Chen H, Aït-Ahmed N, Zaïm EH, et al. (2012) Marine tidal current systems: State of the art. *IEEE Int Symp Ind Electron*, 1431–1437. https://doi.org/10.1109/ISIE.2012.6237301
- 100. Yin X, Zhao X (2021) Optimal power extraction of a two-stage tidal turbine system based on backstepping disturbance rejection control. *Int J Electr Power Energy Syst* 132: 107158. https://doi.org/10.1016/j.ijepes.2021.107158
- 101. Knight C, Mcgarry S, Hayward J, et al. (2014) A review of ocean energy converters, with an Australian focus. 2: 295–320. https://doi.org/10.3934/energy.2014.3.295
- 102. Touimi K, Benbouzid M, Tavner P (2018) Tidal stream turbines: With or without a Gearbox? *Ocean Eng* 170: 74–88. https://doi.org/10.1016/j.oceaneng.2018.10.013
- 103. Elzalabani M, Fahmy FH, Nafeh AESA, et al. (2015) Modelling and simulation of tidal current turbine with permanent magnet synchronous generator. *TELKOMNIKA Indones J Electr Eng*, 13. https://doi.org/10.11591/telkomnika.v13i1.7017
- 104. Tian W, Mao Z, Ding H (2018) Design, test and numerical simulation of a low-speed horizontal axis hydrokinetic turbine. Int J Nav Archit Ocean Eng 10: 782–793. https://doi.org/10.1016/j.ijnaoe.2017.10.006
- 105. Touimi K (2020) Design optimization of a gearbox driven tidal stream turbine. *Universite de Bretagne Occidentale*. https://doi.org/10.1016/j.oceaneng.2018.10.013
- 106. Habibi H, Nohooji HR, Howard I (2017) Power maximization of variable-speed variable-pitch wind turbines using passive adaptive neural fault tolerant control. 12: 377–388. https://doi.org/10.1007/s11465-017-0431-4
- 107. Njiri JG, Söffker D (2016) State-of-the-art in wind turbine control: Trends and challenges. *Renewable Sustainable Energy Rev* 60: 377–393. https://doi.org/10.1016/j.rser.2016.01.110
- 108. Bensalah A, Benhamida MA, Barakat G, et al. (2018) Large wind turbine generators: State-ofthe-art review. 2018 XIII Int. Conf Electr Mach, 2205–2211. https://doi.org/10.1109/ICELMACH.2018.8507165
- 109. Rourke FO, Boyle F, Reynolds A (2010) Marine current energy devices: Current status and possible future applications in Ireland. *Renewable Sustainable Energy Rev* 14: 1026–1036. https://doi.org/10.1016/j.rser.2009.11.012
- 110. Zhou Z, Benbouzid M, Charpentier J, et al. (2017) Developments in large marine current turbine technologies—A review. *Renewable Sustainable Energy Rev* 71: 852–858. https://doi.org/10.1016/j.rser.2016.12.113
- 111. Hu C, Tang C, Yuwen C, et al. (2021) Coupled interactions analysis of a floating tidal current power station in uniform flow. *J Mar Sci Eng*, 9. https://doi.org/10.3390/jmse9090958
- 112. Chen H, Li Q, Benbouzid M, et al. (2021) Development and research status of tidal current power generation systems in China. *J Mar Sci Eng*, 9. https://doi.org/10.3390/jmse9111286
- 113. Si Y, Liu X, Wang T, et al. (2022) State-of-the-art review and future trends of development of tidal current energy converters in China. *Renewable Sustainable Energy Rev* 167: 112720. https://doi.org/10.1016/j.rser.2022.112720

- 114. Zhang Y, Shek JKH, Mueller MA (2023) Controller design for a tidal turbine array, considering Renewable 216: 119063. both power and loads aspects. Energy https://doi.org/10.1016/j.renene.2023.119063
- 115. Lamy JV, Azevedo IL (2020) Do tidal stream energy projects o ff er more value than offshore wind farms? A case study in the United Kingdom. Energy Policy 113: 28-40. https://doi.org/10.1016/j.enpol.2017.10.030
- 116. Ma Y, Hu C, Li L (2021) Hydrodynamics and wake flow analysis of a Π-type vertical axis twintidal rotor current turbine in surge motion. Ocean Eng 224: 10862. https://doi.org/10.1016/j.oceaneng.2021.108625
- 117. Hussain A, Arif SM, Aslam M (2017) Emerging renewable and sustainable energy technologies: State of the Renewable Sustainable Energy Rev 71: 12 - 28. art. https://doi.org/10.1016/j.rser.2016.12.033
- 118. Tekobon J, Chabour F, Nichita C (2017) Development of HILS emulator for a hybrid windpower system. Proc 2016 Int Conf Electr Sci Technol tidal Maghreb, 1 - 8. https://doi.org/10.1109/CISTEM.2016.8066821
- 119. Lande-sudall D, Stallard T, Stansby P (2018) Co-located offshore wind and tidal stream turbines: of energy yield and loading. Renewable 118: 627-643. Assessment Energy https://doi.org/10.1016/j.renene.2017.10.063
- 120. Nkwanyana TB, Siti MW, Wang Z, et al. (2023) An assessment of hybrid-energy storage systems environments. in the renewable JEnergy Storage 72: 108307. https://doi.org/10.1016/j.est.2023.108307
- 121. Kanagji L, Raji A, Orumwense E (2024) Optimizing sustainability offshore hybrid tidal-wind energy storage systems for an off-grid coastal city in South Africa. Sustainability 16: 913. https://doi.org/10.3390/su16219139
- 122. Caraiman G, Nichita C, Mînzu V, et al. (2011) Concept study of offshore wind and tidal hybrid conversion based on real time simulation. Renewable Energy Power Qual J 1: 812-817. https://doi.org/10.24084/repqj09.459
- 123. Lande-sudall D, Stallard T, Stansby P (2019) Co-located deployment of o ff shore wind turbines with tidal stream turbine arrays for improved cost of electricity generation. Renewable Sustainable Energy Rev 104: 492–503. https://doi.org/10.1016/j.rser.2019.01.035
- 124. Faridnia N, Habibi D, Lachowicz S, et al. (2019) Optimal scheduling in a microgrid with a tidal generation. Energy 171: 435-443. https://doi.org/10.1016/j.energy.2018.12.079
- 125. Rahman ML, Nishimura K, Motobayashi K, et al. (2014) Characteristic of small-scale BESS for HOTT generation system. Dig Tech Pap-InnoTek 2014 2014 IEEE Innov Technol Conf, 1-9. https://doi.org/10.1109/InnoTek.2014.6877363



© 2025 the Author(s), licensee AIMS Press. This is an open access AIMS Press article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0)